

GOPEN ACCESS

Citation: Baker JR, Silove D, Horswood D, Al-Shammari A, Mohsin M, Rees S, et al. (2021) Psychological distress, resettlement stress, and lower school engagement among Arabic-speaking refugee parents in Sydney, Australia: A crosssectional cohort study. PLoS Med 18(7): e1003512. https://doi.org/10.1371/journal. pmed.1003512

Received: October 4, 2019

Accepted: June 17, 2021

Published: July 12, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pmed.1003512

Copyright: © 2021 Baker et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data that support the findings of this study will be made available to researchers who meet the criteria for access to study data, after permission from the relevant ethics body below has been sought. The Human RESEARCH ARTICLE

Psychological distress, resettlement stress, and lower school engagement among Arabicspeaking refugee parents in Sydney, Australia: A cross-sectional cohort study

Jess R. Baker[®]*, Derrick Silove, Deserae Horswood[®], Afaf Al-Shammari[®], Mohammed Mohsin[®], Susan Rees[®], Valsamma Eapen[®]

School of Psychiatry, University of New South Wales, Sydney, Australia

* Jessica.baker@unsw.edu.au

Abstract

Background

Schools play a key role in supporting the well-being and resettlement of refugee children, and parental engagement with the school may be a critical factor in the process. Many resettlement countries have policies in place to support refugee parents' engagement with their children's school. However, the impact of these programs lacks systematic evaluation. This study first aimed to validate self-report measures of parental school engagement developed specifically for the refugee context, and second, to identify parent characteristics associated with school engagement, so as to help tailor support to families most in need.

Methods and findings

The report utilises 2016 baseline data of a cohort study of 233 Arabic-speaking parents (77% response rate) of 10- to 12-year-old schoolchildren from refugee backgrounds across 5 schools in Sydney, Australia. Most participants were born in Iraq (81%) or Syria (11%), and only 25% spoke English well to very well. Participants' mean age was 40 years old, and 83% were female. Confirmatory factor analyses were run on provisional item sets identified from a literature review and separate qualitative study. The findings informed the development of 4 self-report tools assessing parent engagement with the school and school community, school belonging, and quality of the relationship with the schools' bilingual cultural broker. Cronbach alpha and Pearson correlations with an established Teacher–Home Communication subscale demonstrated adequate reliability ($\alpha = 0.67$ to 0.80) and construct and convergent validity of the measures (p < 0.01), respectively.

Parent characteristics were entered into respective least absolute shrinkage and selection operator (LASSO) regression analyses. The degree of parents' psychological distress (as measured by the Kessler10 self-report instrument) and postmigration living difficulties (PLMDs) were each associated with lower school engagement and belonging, whereas less time lived in Australia, lower education levels, and an unemployed status were associated with higher ratings in relationship quality with the schools' cultural broker. Study limitations Ethics Coordinator The University of New South Wales Email: humanethics@unsw.edu.au.

Funding: This research was part funded by the National Health and Medical Research Council of Australia (NHMRC) program grant (APP1073041) awarded to DS (CI) and SR (AI). The NHMRC had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript declare that no competing interests exists.

Abbreviations: CALD, culturally and linguistically diverse; CBRS, Cultural Broker Relationship Scale; CFA, confirmatory factor analysis; CFI, comparative fit index; EAD/L, English as an Additional Language or Dialect; HTQ, Harvard Trauma Questionnaire; K10, Kessler Psychological Distress Scale; LASSO, least absolute shrinkage and selection operator; PMLD, postmigration living difficulty; PRESS, predicted residual sum of squares; PSSM, Psychological Sense of School Membership; PTE, potentially traumatic event; SBS-RP, School Belonging Scale-Refugee Parent; SCES-RP, School Community Engagement Scale-Refugee Parent; SIES-RP, School Internal Engagement Scale-Refugee Parent; SRMR, standardised root mean square residual; TE, traumatic event; THC, Teacher-Home Communication; TLI, Tucker-Lewis index; VIF, variance inflation factor.

include the cross-sectional design and the modest amount of variance (8% to 22%) accounted for by the regression models.

Conclusions

The study offers preliminary refugee-specific measures of parental school engagement. It is expected they will provide a resource for evaluating efforts to support the integration of refugee families into schools. The findings support the need for initiatives that identify and support parents with school-attending children from refugee backgrounds who are experiencing psychological distress or resettlement stressors. At the school level, the findings suggest that cultural brokers may be effective in targeting newly arrived families.

Author summary

Why was this study done?

- Current measures of parent school engagement do not consider the unique experiences of families from refugee backgrounds.
- This study was undertaken to develop appropriate measures of refugee parents' engagement with school.
- It is expected that specific measures to ascertain refugee parents' engagement will help support the development and evaluation of school-based refugee programs.

What did the researchers do and find?

- Using data from a cohort study of 233 Arabic-speaking refugee families, the study developed and validated 4 self-reports tools to assess parent school engagement and their sense of belonging.
- The study also explored the quality of the relationship between parents and the bilingual officer employed by schools to act as a cultural broker.
- Our findings suggest that parents who were experiencing psychological distress or postmigration living difficulties (PLMDs) were less engaged with their child's school.

What do these findings mean?

- These new refugee-specific measures of parent engagement are expected to assist in evaluating programs aimed at supporting the integration of refugee parents and their children into schools.
- The cross-sectional design precludes causal interpretations.
- Nonetheless, the findings highlight a need to consider the identification and capacity building of parents who are experiencing psychological distress or resettlement stressors.



Introduction

Global refugee numbers are at a record high, of which school aged children make up over 52% [1]. In high-income countries of resettlement, school attendance is mandatory. This provides an opportunity to engage with parents from refugee backgrounds, in order to promote the well-being and community integration of refugee families as a whole [2–4]. In that regard, refugee parents' relationship with the school may be a critical factor in the successful adaptation of refugee students [5,6]. Robust evidence suggests that across ethnic groups and sociodemographic backgrounds, parent engagement with schools is positively correlated with children's well-being. More specifically, children whose parents participate in school activities are more engaged in their learning and more successful in terms of achievement, attendance, motivation, school completion, emotional adjustment, prosocial behaviour, and peer interactions [7–14].

Several countries such as the United Kingdom, United States of America, Canada, Australia, Sweden, and Germany have government policies and programs in place to support the integration of refugee families into schools. Targeted measures that capture the distinct ways in which refugee parents might engage with schools are needed to monitor the success of these initiatives. Specialised measures may also generate data to better advance efforts to improve refugee parents' engagement in their children's schooling in a tailored way that is most valuable to the family. Teasing out the nuances of refugee parent school engagement relative to non-refugee families has the additional benefit of avoiding any cultural biases, including misguided findings that refugee families might be less invested in their children's education. Several theories of parent school engagement have been proposed [15-17]. The most common and comprehensive framework proposed is Epstein's 6 different types of parent engagement [16]. The first type of engagement is parenting and discusses the school's role in educating families on how to create a positive home environment. The second type of engagement is centered on effective communication between home and school. The other types include recruiting parents to volunteer at the school, involving parents in school decision-making, home learning, and community collaborations.

These traditional definitions of parental school involvement may not adequately capture engagement as it applies to refugee families [18]. Refugee parents' limited English proficiency [19–23] may interfere with volunteering at the school and supporting their child's learning at home. This issue may be exacerbated by time constraints related to work, family, and economic survival [19,20,23,24]. Disruptions in schooling during displacement and limited family experience with school systems in countries of resettlement may also challenge typical parent engagement [24,25]. Qualitative research also suggests cultural dissonance between expectations of parents and the school. For instance, culture of origin may influence refugee parents' belief that it is inappropriate to "interfere" with the work of teachers and that keeping one's distance may be deemed to show respect for the school and its staff [26].

Indeed, the high value refugee parents place on their child's education may be expressed in different ways [23]. While not focused on refugee children, a large national study of students in the US found that academic achievement among white American children was positively associated with parent engagement in school-based activities; however, for Asian American families, the relationship was inverse [27]. In another study, ethnicity was found to influence the relationship between different types of parent engagement categorised within Epstein's typology and student achievement [28]. For instance, volunteering was a better predictor of achievement for white students than for Asian, Hispanic, and black students, while contact with the school was more important for explaining achievement effects for Hispanic and white students than for Asian and black students.

Moreover, an emphasis on behavioural indicators of parent engagement [15–17] may undervalue the relational or more affective components of parental engagement, such as how close or connected a parent feels to the school or school community. The well-being of refugee parents is affected by historical factors such as past trauma and displacement [29] and learned distrust of authorities based on exposure to persecution and discrimination [30]. Establishing parental trust and safety may therefore be foundational to effective engagement with schools. For example, in a study of Hmong refugees in the US, a school ethos of inclusion was found to assist parents to overcome expectations of deference towards "authority" [23].

Numerous case studies have showcased how adept certain schools and their staff have become at engaging with refugee families [31,32]. Current measures of parent school engagement may not capture these specific activities attuned to refugee families [33]. For instance, schools have been found to play an instrumental role in guiding parents to local support agencies and helping them navigate unfamiliar and often overwhelming government systems [31,34]. Similarly, reports of social activities at the school, such as breakfast clubs and diversity days, have been posited to connect parents with each other, while school-initiated parent classes to support acculturation in the region's primary language have been documented to improve cultural competency [35]. A key piece in the literature is the consistent role that bilingual school liaison officers play in facilitating these processes [6,23,26,36]. These persons are employed by the school, specifically tasked with brokering the relationship between refugee families and the school, as well as the wider community. To date, no measure exists to capture the quality of the relationship between the cultural broker and parents.

The primary aim of the present study was to develop and validate self-report tools to measure refugee parents' engagement with school. We draw on Epstein's framework and extend it to include specific elements of engagement that foster families' social and cultural capital [31,32]. We also aimed to develop and validate a novel self-report tool that includes an affective index of engagement—namely parents' sense of belonging or feelings of connectedness to the school. Further, we aimed to include an assessment of the relationship between refugee parents and the schools' cultural broker. An established measure of teacher–home communication, a robust facet of parent engagement, was selected as a distal measure of convergent validity [37– 39].

The secondary aim of the present study was to identify refugee parent characteristics associated with these new measures of parent engagement. While some refugee families may engage readily and easily with school and not require special assistance, other families may have the opposite tendency and require more active assistance. Identifying refugee parents at "risk" of low levels of engagement can help schools to individualise or tailor their support to families most in need. It was reasoned that these factors would likely mirror some of the barriers to traditional engagement described earlier [19–24]. For instance, in the migrant literature, low socioeconomic status, less education, and time preoccupation with daily environmental stressors have been associated with lower parent engagement [40–46].

Specifically, it was hypothesised that countervailing parental factors such as length of time lived in the host country, employment and visa status, psychological health, English language proficiency, trauma experience, and postmigration stressors would be associated with school engagement and belonging.

Method

Design

The present report is based on the baseline cross-sectional data of a longitudinal cohort study exploring school climate and refugee student well-being.



4/22

Ethics statement

The study was approved by the University's Human Research Ethics Committee (HC15833) and the equivalent body of the State Department of Education (SERAP 2016056) (see <u>S1 Prospective Protocol</u>). The bilingual research team was trained in working with refugee families using sensitive interviewing techniques and received ongoing supervision by author JB throughout the study. The principles governing recruitment were implemented according to the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guide-lines (see <u>S1 STROBE Checklist</u>).

Setting

The study was conducted at 5 primary schools in a low socioeconomic multicultural geographical area of Sydney, Australia in 2016. The urban region receives a substantial portion of Australia's refugee intake each year. For example, over 12% of Australia's 73,833 refugee intake was settled in the catchment area between 2010 and 2015. An additional 12,000 Syrians were also resettled as a special intake in the area between 2016 and 2017 [47]. Given several years of experience working with refugee families, the schools had many well-established support initiatives, such as new family orientations, small support classes for refugee students, events to celebrate cultural diversity, homework clubs, English language and computer literacy classes for parents, and parent cafes and breakfast clubs.

Participants

Refugees speaking Arabic were the largest group entering the country at the time. Confining the study to one participant language limited the risk of transcultural measurement error and small cell sizes that would eventuate if multiple ethnic groups were included. Participants were Arabic-speaking parents of Year 5 and 6 students comprising 10- to 12-year-old children, who entered Australia on a humanitarian (refugee) visa, and had attended the school for at least 3 months. Selection criteria permitted either mother or father to participate at the family's choosing. Assuming that the maximum standard deviation would not exceed 15 for the largest subscale total score for the project, and applying a 95% confidence interval along with a 3% margin of error, a sample size of 96 was deemed sufficient to reject the null hypothesis at the 0.05 probability level.

A nominated bilingual cultural broker at the school invited eligible families to participate, passing on the contact details of assenting families to the bilingual field team. At the point of further contact, the study process and objectives were explained fully in Arabic and, if the parent provided written informed consent, the family was included in the study (see flow diagram in Fig 1). Families received a \$30 gift voucher for their participation. The design involved a rolling recruitment strategy of all eligible students attending or joining the school over a 2-year time frame, with the largest intake occurring at the start of each school year.

Measures

Participants completed an extensive demographic survey followed by a provisional set of items for the school engagement measures (see a description hereafter), and then lastly, Arabic versions of the established measures below. All translations were checked for local cultural appropriateness and accuracy. To circumvent any literacy or comprehension issues, the measures were completed in an interview style, with the questions read aloud to participants in Arabic.





Fig 1. Flow diagram of parent participant recruitment.

https://doi.org/10.1371/journal.pmed.1003512.g001

Item development and pilot for new measures

Items were developed from 2 main sources: (i) a desk review of the refugee school literature and existing school engagement scales [48,49]; and (ii) themes emerging from a preliminary qualitative study exploring the school experiences of refugee families (to be published elsewhere). The informing qualitative study included two 10 to 12 year olds (and their parents) from refugee backgrounds from each of the 5 schools of the present study, plus one other school that opted out before the quantitative stage of the current study (n = 12 dyads). In addition, focus groups were held with staff at each school. These typically comprised a Year 6 Teacher, School Counsellor, an English as an Additional Language or Dialect (EAD/L) Teacher, and one bilingual cultural broker. In total, 26 staff members participated.

Based on the desk review and qualitative study, authors JB and DH generated a pool of potential items written in English aimed at measuring school engagement. These were reviewed and edited for content and cultural appropriateness by a local primary school teacher and author AA (an active member of the local Arabic-speaking community with excellent English language skills). Author AA translated the items into Arabic, which were assessed for comprehension and cultural congruence by Arabic members of the research team. The items were piloted with sequential parents enrolled in the study. Items were deleted if they proved difficult to understand, required excessive explanation, or were interpreted differently by parents. This iterative process, involving feedback and research group discussion, led to a final consensus regarding a provisional set of school engagement items, a provisional set of items pertaining to the cultural broker–parent relationship, and a provisional set of items can be found in supporting file Table A in <u>S1 Data Tables</u>.

Refugee parent school engagement items

The 23-item measure followed 5 themes identified in the preliminary qualitative study (to be published elsewhere). The themes were (i) ease of communication between the school and parent (5 items, e.g., "It would be difficult for me to find a way to discuss an issue with school staff"); (ii) practical assistance provided by the school such as accessing services or provision of food or uniforms (5 items, e.g., "School staff direct me to services I need or that might be



helpful to access"); (iii) degree of parent participation in school activities (6 items, e.g., "I regularly attend school events"); and the ways in which the school facilitates (iv) social capital (4 items, e.g., "Through the school, I have met and made friends"); or (v) acculturation (3 items, e.g., "This school helps me feel like I belong in Australia"). To maintain consistency with the Teacher–Home Communication (THC) scale, items were rated on a 4-point Likert scale from 1 (disagree a lot) to 4 (agree a lot). Negatively worded items were reverse scored. Higher scores reflected greater school engagement.

Cultural broker-parent relationship items

Six items pertained to the relationship between the parent and an identified bilingual cultural broker at the school employed to support newly arrived culturally and linguistically diverse (CALD) families in the transition to school and foster engagement between the school and broader CALD community [6,23,26]. Example items include: "XX has been important in making me feel secure in the school because xx understands my family's religious and cultural backgrounds" and "XX helps me with my family's wellbeing." Items were rated on a 4-item Likert scale. For parents unable to identify a cultural broker, items were coded as "Not Applicable" and excluded from any analysis. Higher scores reflected a higher-quality relationship with the cultural broker.

Refugee parents' sense of school belonging items

Distinct to the measures above, parent sense of school belongingness items were based on an adaptation of the well-established student-rated Psychological Sense of School Membership (PSSM) scale [50]. Items measured feelings of belonging and engagement with the school and were rephrased to ensure relevance to parents (e.g., "I am treated with as much respect as other parents in the school"). Two irrelevant items concerning school performance were excluded. The resulting 16-item scale was rated and scored in accordance with the PSSM scoring procedure on a 5-point Likert scale (1 "Not at all true" to 5 "Completely true"). Negatively worded items were reverse scored. Higher scores reflected a greater sense of belonging.

Established measures

Teacher–Home Communication subscale of the Delaware School Climate Survey. The Delaware School Climate Survey measures parent perceptions of the "quality and character of school life" across several domains utilising a 4-point Likert scale (1 "Disagree a lot" to 4 "Agree a lot"). For the study purpose, the 4-item THC subscale was summed, with higher scores reflecting more positive communication. The parent survey has shown excellent internal consistency across ethnicity, sex, and grade-level groups, and correlations with academic achievement, bullying victimisation, and school suspensions demonstrate concurrent validity [52]. The current sample demonstrated adequate internal consistency for the THC (Cronbach $\alpha = 0.69$).

Kessler Psychological Distress Scale (K10). The K10 is a 10-item self-report measure of nonspecific psychological distress for adults (e.g., "How often did you feel nervous?"). Participants rated their experience of distress in the preceding 30 days on a 5-point Likert scale (1 "None of the time" to 5 "All of the time"). Items were added to produce a total score; the higher the score, the greater the level of distress. Arabic NAATI-accredited translations were obtained from the state government health body. Normative data have been established for Australia [53], and high internal consistency demonstrated with refugee groups ($\alpha = 0.86$) [54]. The total score obtained from the present sample demonstrated excellent internal consistency (Cronbach $\alpha = 0.89$).



Postmigration Living Difficulties Scale (PMLD). The PMLD scale consisted of 28 items reflecting levels of distress caused by daily living difficulties related to postmigration adaptation over the previous 12 months (e.g., "separation from family" or "delays processing your visa application"). Items are rated on a 5-point scale (0: "not a problem" to 4: "a very serious problem"). The scale has consistently been identified as a predictor of psychopathology among displaced populations [55]. The Arabic translation was obtained with permission from the lead author of the scale [56]. The total PMLD counts were summed to create a continuous total score. Internal reliability for PMLD items for the study sample was 0.79.

The Harvard Trauma Questionnaire (HTQ)—Iraqi version. The HTQ is the most widely used measure of trauma exposure and posttraumatic stress disorder in the refugee field [57,58]. The initial section includes a checklist of potentially traumatic events (PTEs), including exposure to combat, forms of warfare, and separations and losses. The list was modified to reflect the experiences of the refugee populations being studied. Participants reported on their lifetime exposure to 23 events (Yes/No). Items were coded "1" for yes, and "0" for no for lifetime exposure. The items were summed to generate a total trauma count ranging from 0 to 23.

Data analysis

Refinement and psychometric testing of the new measures. Confirmatory factor analysis (CFA) using MPLUS 7.1 was conducted on the 3 provisional measures above, with the aim to identify component scales and relevant item loadings. The threshold was set at p < 0.05. Items with lowest loadings were deleted serially to test for improved model fit based on a composite comparative fit index (CFI), Tucker–Lewis index (TLI), and standardised root mean square residual (SRMR) [51]. The final sets of items were aggregated, and transformations used to generate mean scores.

Reliability and validity tests were conducted using IBM SPSS Statistics Version 25 [59] and MPLUS 7.1 [51]. For reliability estimates, Cronbach alpha (α) was calculated for each scale. Positive correlations between items within each scale and with the single total score as indicated in a CFA were used to indicate construct validity. As an indicator of convergent validity, the data were examined for positive correlations among each derived scale, as well as the THC subscale.

Predictive analyses with developed measures. Firstly, descriptive statistics were presented for all parental level characteristics. Next, variance inflation factor (VIF) and tolerance statistics examined the multicollinearity among predictors accounting for each outcome factor. The original intention was to conduct linear or stepwise regression analyses. However, this was changed at the peer review stage, to LASSO (forward) regression analyses. Accordingly, parental level characteristics were entered into separate LASSO (forward) regression analyses with each newly developed scale as the outcome variable. The LASSO regressions were performed with SAS 9.4 software, PROC GLMSELECT with selection = forward (select = SL stop = predicted residual sum of squares (PRESS). To adjust for school effects, school was entered as a predictor variable. Based on the 5 sampled schools A, B, C, D, and E (designated letters for confidentiality), 4 indicators were created considering the fifth school (school E) as the reference group. Differences in measurement units of predictor variables meant that standardised LASSO regression coefficients were presented.

Results

Sample characteristics

The anticipated resettlement of 12,000 refugee families into the local area was delayed, thus the sample size was lower than prospective estimates. Of the 303 families identified by the schools



to take part, 283 families assented to be contacted by the research team, and a final 233 parents (77% of those identified) consented to participate—see flow diagram of parent participant recruitment in Fig 1. Parent characteristics are provided in Table 1. Most participants were married women, born in Iraq (consistent with Australia's refugee intake at the time), and spoke little or no English. Around a third were educated beyond high school, and few participants were currently employed. Around half met criteria for a mental health disorder based on the K10, and almost all endorsed experiencing at least one traumatic event (TE).

Confirmatory factor analysis

Please see Fig 2 showing the process of measure development.

Refugee parents school engagement

The CFA examined the correlations between the 5 predefined themes of communication between home and school, practical assistance provided by the school, parent participation in school activities, school as facilitating social capital, and school as enabling acculturation. The dimensions of communication, assistance, parent participation, and social capital correlated significantly with each other (r = 0.25 to 0.96, *p*-values ranged from 0.037 to 0.001). Acculturation did not show a correlation with the communication or assistance dimensions (p > 0.0.05) but significantly correlated with the social capital dimension (r = 0.68, p < 0.001) (see Table B in S1 Data Tables). Based on these findings, the items were aggregated into 2 notional and statistically distinct scales: (i) refugee parent engagement in relation to communication, assistance, and participation within the internal school system—named the School Internal Engagement Scale-Refugee Parent (SIES-RP) (16 items); and (ii) broader refugee parent engagement within the school community in which the school facilitates social capital and acculturation—named the School Community Engagement Scale-Refugee Parent (SCES-RP) (7 items). A CFA was performed on each separate scale. After item reduction as described above, the final CFA model fitness indices are presented in Table 2. Eleven of the original 16 items were retained for the SIES-RP, while all 7 items of the SCES-RP were retained (see Tables C and D in S1 Data Tables, respectively).

Cultural broker relationship

Final CFA model fitness indices are presented in <u>Table 2</u>. Five of the original 6 items were retained. This was named the Cultural Broker Relationship Scale (CBRS) (see Table E in <u>S1</u> Data Tables).

Refugee parents' sense of school belonging

Final CFA model fitness indices are presented in <u>Table 2</u>. Fifteen of the original 16 items were retained. This was named the School Belonging Scale-Refugee Parent (SBS-RP) (see Table F in <u>S1 Data Tables</u>).

Final measures

In summary, the refinement procedures produced 4 scales: the SIES-RP, the SCES-RP, the CBRS, and the SBS-RP. Cronbach α and other summary statistics are shown in Table 2.

Reliability and validity testing of developed measures

Tables detailing the construct validity of the final developed measures are presented in Tables G-J in S1 Data Tables. Descriptive characteristics and Cronbach α (i.e., reliability) of



Parents' characteristics	Number	% of total
All#	233	100.0
Sex		
Male	39	16.7
Female	194	83.3
Parents' age		
<35 years	62	27.0
35-44	108	47.0
45 and older	60	26.0
Mean (SD)	39.7(6.8)	
Marital status		
Married	201	86.3
Divorced/Separated/Widowed	32	13.7
Current parent visa category in Australia		
Humanitarian refugee/Family sponsored	178	76.4
Permanent resident or Citizen	55	23.6
Duration of living in Australia		
Under 2 years	66	30.1
2 to 5 years	90	41.1
More than 5 years	63	28.8
Mean (SD)	3.8(3.0)	
Country of birth		
Iraq	189	81.1
Syria	26	11.2
Other	18	7.7
Spoken language at home		
Arabic	161	69.1
Others	72	30.9
English proficiency		
A little/Not at all	174	74.7
Well/Very Well	59	25.3
Highest level of education attained		
Up to high school	152	65.2
Diploma and University degree	81	34.8
Employment status		
Unemployed and others	216	92.7
Employed	17	7.3
Family income (annual)		
up to \$37,000	79	33.9
\$37,001 and above	154	66.1
Parent received any counselling or psychological therapies in Australia		
No counselling or therapy	129	58.1
NGO service	55	24.8
GP	10	4.5
Private practice psychologist/counsellor	25	11.3
Others	3	1.4
Number TEs experienced: Mean (SD)	9.9(4.0)	
Number of PMLDs: Mean (SD)	5.4(0.3)	

Table 1. Parent characteristics.

(Continued)



Table 1. (Continued)

Parents' characteristics	Number	% of total
Psychological distress categories (measured as K10) : Mean (SD)	21.3(9.9)	
Child's enrolled school		
School A	46	19.7
School B	43	18.5
School C	60	25.7
School D	33	14.2
School E	51	21.9

[#]The numbers do not always add up to 233 due to missing data. The number of missing observations for "Parents duration of living in Australia" was 14 (6%) and for "Parent received any counselling or psychological therapies in Australia" was 11 (4.7%). Else, the number of missing cases ranges from 0 to 3 (0 to 1.2).

GP, general practitioner; K10, Kessler Psychological Distress Scale; NGO, nongovernment organisation; PMLD, postmigration living difficulty; TE, traumatic event.

https://doi.org/10.1371/journal.pmed.1003512.t001

the new measures, as well as their correlations with THC, are provided in Tables $\underline{3}$ and $\underline{4}$, respectively. All measures were significantly correlated with THC indicating convergent validity. The SBS-RP alpha score also indicated very good reliability. The SCES-RP and SBS-RP reported alpha scores of a minimally acceptable standard. The Cronbach α of the SIES-RP and CBRS did not meet the typically acceptable standard of 0.70, reported as 0.67 and 0.63, respectively.



Fig 2. The process of measure development.

https://doi.org/10.1371/journal.pmed.1003512.g002



		Model summary					
Scales measures	CFI	TLI	SRMR	AIC	Sample size-adjusted BIC	Cronbach α	
SIES-RP							
Model 1 (16 items)	0.47	0.38	0.09	9,971.0	9,984.1	0.63	
Model 2 (11 items)	0.60	0.49	0.08	6,834.6	6,843.6	0.67	
SCES-RP							
Model 1 (7 items)	0.84	0.75	0.06	4,519.9	4,525.6	0.73	
SBS-RP							
Model 1 (16 items)	0.74	0.70	0.07	8,185.7	8,199.0	0.77	
Model 2 (15 items)	0.76	0.72	0.07	7,435.3	7,447.7	0.80	
CBRS							
Model 1 (6 items)	0.79	0.65	0.08	2,820.5	2,824.3	0.53	
Model 2 (5 items)	0.81	0.62	0.08	2,166.1	2,169.2	0.63	

Table 2. Values of CFI, TLI, SRMR, AIC, sample size-adjusted BIC, and Cronbach α for each CFA model with number of items included in CFA.

AIC, Akaike information criterion; BIC, Bayesian information criterion; CBRS, Cultural Broker Relationship Scale; CFA, confirmatory factor analysis; CFI, comparative fit index; SBS-RP, School Belonging Scale-Refugee Parent; SCES-RP, School Community Engagement Scale-Refugee Parent; SIES-RP, School Internal Engagement Scale-Refugee Parent; SRMR, standardised root mean square residual; TLI, Tucker–Lewis index.

https://doi.org/10.1371/journal.pmed.1003512.t002

Predictive analyses with developed measures

The mean of the school measures by parent characteristics are presented in Table 5. Results of the standardised LASSO regression analyses coefficients (Beta, β) for significant predictors are presented in Table 6. Table 7 presents the forward LASSO regression selection summary for each outcome variable. The VIF for the predictor variables ranged from 1.14 to 2.25. Multicol-linearity tolerance statistics ranged from 0.44 to 0.87. While some predictor variables were significantly correlated with one another, the maximum value of correlation coefficient r = 0.48 was less than <0.70; the maximum VIF value of 2.25 was less than 10, and the minimum tolerance statistic of 0.44 was greater than 0.10. This indicated that multicollinearity among predictors was sufficiently low to warrant entry into the LASSO regression.

School Internal Engagement Scale for Refugee Parents (SIES-RP)

Participants with children at schools C and D reported significantly greater internal school engagement as compared to school E (reference school). The LASSO regression analysis revealed that greater parent psychological distress and the longer the time lived in Australia

Measures (number of items in the scale and sample size)	Mean total scale score (SD) [range]	Mean of scale score (SD) [range]	Cronbach α
SIES-RP [11 items] $(n = 231)$	32.6 (5.4) [16-44]	2.9 (0.5) [1-4]	0.67
SCES-RP [7 items] $(n = 231)$	18.1 (4.7) [7–28]	2.6 (0.7) [1-4]	0.73
SBS-RP [15 items] (<i>n</i> = 232)	66.7 (6.5) [41–77]	4.4 (0.4) [1-5]	0.80
CBRS [5 items] (<i>n</i> = 216)	17.6 (2.5) [5-20]	3.5 (0.5) [1-4]	0.63
THC subscale [4 items, parent report] (<i>n</i> = 233)	14.7 (1.6) [7–16]	3.7 (0.4) [1–4]	0.69

Table 3. School scale measures with mean scale score and reliability coefficients Cronbach a.

CBRS, Cultural Broker Relationship Scale; SBS-RP, School Belonging Scale-Refugee Parent; SCES-RP, School Community Engagement Scale-Refugee Parent; SIES-RP, School Internal Engagement Scale-Refugee Parent; THC, Teacher–Home Communication.

https://doi.org/10.1371/journal.pmed.1003512.t003



Table 4. Correlation matrix of among scale school measures.

Measures (number of items in the scale and sample size)	SIES-RP	SCES-RP	SBS-RP	CBRS	THC
SIES-RP	1.00				
SCES-RP	0.30**	1.00			
SBS-RP	0.53**	0.22**	1.00		
CBRS	0.16*	0.01	0.28**	1.00	
THC scale	0.51**	0.23**	0.60**	0.19**	1.00

*Correlation significant at p < 0.05.

**Significant at p < 0.01.

CBRS, Cultural Broker Relationship Scale; SBS-RP, School Belonging Scale-Refugee Parent; SCES-RP, School Community Engagement Scale-Refugee Parent; SIES-RP, School Internal Engagement Scale-Refugee Parent; THC, Teacher–Home Communication.

https://doi.org/10.1371/journal.pmed.1003512.t004

were each related to lower internal school engagement. Parents on a permanent residency or citizenship visa (as opposed to a humanitarian visa) reported greater internal school engagement. Length of time lived in Australia and parent visa category exerted the largest influence on parents' school engagement, followed by the index of psychological distress. The overall model was significant, accounting for 22.3% of the total variance.

School Community Engagement Scale for Refugee Parents (SCES-RP)

The LASSO regression analysis revealed that speaking a language other than Arabic, being unemployed, and endorsing more PMLDs were significantly associated with lower school community engagement. The overall model accounted for 9.7% of the total variance.

School Belonging Scale-Refugee Parent (SBS-RP)

The regression analysis revealed that lower level of education, more PMLDs, and greater psychological distress were significantly correlated with lower school belonging. School D had significant influence on parents' higher school belonging. The overall model was significant and accounted for 14.7% of the total variance.

Cultural Broker Relationship Scale (CBRS)

The regression analysis revealed that parents who had lived less time in Australia, were educated at a high school or diploma level, or who were unemployed reported a significantly greater quality relationship with the school cultural broker, relative to parents who had lived longer in Australia, educated to a degree level, or who were employed. The overall model was significant and accounted for 11.0% of the total variance.

Discussion

This paper describes the development of 4 contemporary measures of refugee parents' engagement with their child's school, namely the SBS-RP, the CBRS, the SIES-RP, and the SCES-RP. The SBS-RP and SCES-RP demonstrated adequate to good reliability. The reliability of the SIES-RP and CBRS was below the acceptable threshold. All measures indicated sound construct validity, and convergent validity as based on positive correlations with the THC scale. Using these instruments, we studied a cohort of Arabic-speaking refugee parents in Australia. We report that parental factors associated with school engagement included time lived in Australia, psychological distress, PMLDs, visa status, employment status, education level, and speaking Arabic (the dominant language of the sample—69%) at home. Greater psychological



Table 5.	Mean of the	school measure	s by parent	characteristics.
----------	-------------	----------------	-------------	------------------

Parental characteristics	SIES-RP	SCES-RP	SBS-RP	CBRS
	Mean (SD)	Mean (SD)	Mean (SD)	Mean (SD)
All [#]	3.0 (0.5)	2.6 (0.7)	4.4 (0.4)	3.5 (0.5)
Sex				
°Male	3.1 (0.4)	2.6 (0.6)	4.5 (0.4)	3.5 (0.4)
°Female	2.9 (0.5)	2.6 (0.7)	4.4 (0.4)	3.5 (0.5)
Marital status				
°Married	3.0 (0.5)	2.6 (0.7)	4.5 (0.4)	3.5 (0.5)
°Divorced/Separated/Widowed	2.9 (0.5)	2.6 (0.8)	4.4 (0.5)	3.5 (0.5)
Current parent visa category in Australia				
°Humanitarian refugee/Family sponsored	2.9 (0.5)	2.6 (0.7)	4.4 (0.4)	3.5 (0.5)
°Permanent resident or Citizen	3.1 (0.5)	2.6 (0.7)	4.5 (0.4)	3.4 (0.5)
Country of birth				
°Iraq	3.0 (0.5)	2.6 (0.7)	4.4 (0.4)	3.5 (0.4)
°Syria	3.0 (0.5)	2.4 (0.7)	4.5 (0.3)	3.6 (0.7)
°Other	2.8 (0.5)	2.4 (0.7)	4.4 (0.4)	3.3 (0.6)
Spoken language at home				
°Arabic	3.0 (0.5)	2.7 (0.7)	4.4 (0.4)	3.5 (0.5)
°Others	2.9 (0.5)	2.4 (0.6)	4.5 (0.4)	3.6 (0.5)
English proficiency				
°A little/Not at all	2.9 (0.5)	2.6 (0.7)	4.4 (0.4)	3.6 (0.5)
°Well/Very Well	3.0 (0.4)	2.7 (0.7)	4.4 (0.4)	3.3 (0.5)
Highest level of education attained				
°Up to high school	2.9 (0.5)	2.6 (0.6)	4.5 (0.4)	3.6 (0.4)
°Diploma and University degree	3.0 (0.5)	2.7 (0.7)	4.4 (0.5)	3.4 (0.6)
Employment status				
°Unemployed and others	3.0 (0.5)	2.6 (0.7)	4.4 (0.4)	3.6 (0.5)
°Employed	3.1 (0.4)	3.0 (0.6)	4.6 (0.3)	3.1 (0.6)
Child's enrolled school				
°School A	2.9 (0.5)	2.6 (0.6)	4.4 (0.5)	3.6 (0.4)
°School B	2.8 (0.5)	2.4 (0.7)	4.5 (0.4)	3.5 (0.7)
°School C	3.1 (0.5)	2.6 (0.7)	4.4 (0.4)	3.5 (0.6)
°School D	3.2 (0.4)	2.7 (0.7)	4.6 (0.3)	3.6 (0.3)
°School E	2.8 (0.5)	2.6 (0.7)	4.4 (0.4)	3.5 (0.5)

CBRS, Cultural Broker Relationship Scale; SBS-RP, School Belonging Scale-Refugee Parent; SCES-RP, School Community Engagement Scale-Refugee Parent; SIES-RP, School Internal Engagement Scale-Refugee Parent.

https://doi.org/10.1371/journal.pmed.1003512.t005

distress was implicated in lower parental school engagement across both internal engagement and the more affective sense of belonging. Another factor to reliably emerge as a predictor of school engagement was PMLDs. Specifically, the greater the number of current living difficulties, the lower the parents' reported community school engagement and sense of school belonging. Interestingly, English language proficiency did not show an association with parent school engagement, cultural broker relationship, or sense of belonging. Regarding the cultural broker scale, parents who had lived less time in Australia, were unemployed, or less educated endorsed a greater quality relationship with this school staff member.

It makes sense that the strongest correlations of the 4 self-report tools were between the SBS-RP and the SIES-RP, the 2 measures assessing sense of belonging and internal school



Table 6. Standardised LASSO (forward) regression coefficients (Beta, β) for significant predictors.

Significant predictors [#]	SIES-RP ($n = 231$)	SCES-RP ($n = 231$)	SBS-RP ($n = 232$)	CBRS (<i>n</i> = 216)
	Beta (β)	Beta (β)	Beta (β)	Beta (β)
Current visa category in Australia	0.26**	-	-	-
Duration in Australia	-0.24**	-		-0.24**
Language spoken at home	-	-0.15*		
Highest level of education completed	-	-	-0.15*	-0.17**
Employment status	-	0.16**	-	-0.14^{*}
Number of PMLDs	-	-0.18**	-0.16*	-
Psychological distress (K10 score)	-0.22**	-	-0.23**	-
School C [#]	0.24**	-	-	-
School D [#]	0.27**		0.16**	
Value of R-square	0.241**	0.118**	0.166**	0.124**
Value of adjusted R-square	0.223**	0.097**	0.147**	0.110**

Predictor variables used in LASSO regression analysis are coded as:

Sex: 0 = Male, 1 = Female;

Age (parent): individual age ranges 26 years to 63 years;

Marital status: 0 = Married, 1 = Divorced/Separated/Widowed;

Country of birth: 0 = Iraq, 1 = Syria and others;

Current visa category in Australia: 0 = Humanitarian/Family-sponsored Refugee, 1 = Permanent Resident/Citizen/Others;

Duration of living in Australia: 0 = Under 2 years, 1 = 2 to 5 years, 3 = More than 5 years;

Language spoken at home (parent): 0 = Arabic, 1 = Others;

Level of English efficiency (parent): 0 = A little/Not at all, 1 = Well/Very Well;

Highest level of education attained: 0 = Up to HSC, 1 = Diploma and University degree;

Employment category: 0 = Unemployed/others, 1 = Employed;

Family income (annual): 0 = up to \$37,000, 1 = \$37,001 and above;

Number of TEs: total counts of premigration traumatic events (0 to 23);

Number of PMLDs: total counts of PMLDs (0 to 25; treated as continuous);

Psychological distress (score): Total score of 10 individual items (K10 score) (10 to 46; treated as continuous);

[#]For 5 enrolled schools, 4 dichotomous dummy variable created: named as School A, School B, School C, School D where School E considered as reference category.

Outcome variables for each LASSO regression analysis are as follows:

SIES-RP subscale score;

SCES-RP subscale score;

SBS-RP subscale score;

CBRS subscale score.

*Significant at p < 0.05.

**Significant at p < 0.01.

CBRS, Cultural Broker Relationship Scale; HSC, xxxx; K10, Kessler Psychological Distress Scale; LASSO, least absolute shrinkage and selection operator; PMLD, postmigration living difficulty; SBS-RP, School Belonging Scale-Refugee Parent; SCES-RP, School Community Engagement Scale-Refugee Parent; SIES-RP, School Internal Engagement Scale-Refugee Parent; TE, traumatic event.

https://doi.org/10.1371/journal.pmed.1003512.t006

engagement (SBS-RP and SIES-RP). The SCES-RP and CBRS encompass community factors outside of the school or independent of the relationship with the teacher or an ability to speak English and thus add more noise to the data.

The finding that parent psychological distress was related to lower internal engagement and sense of belonging also makes sense. A parent facing mental health issues may be less able to engage with their child's school. They may also be harder to engage or feel more disconnected. This latter idea aligns with bidirectional research that family engagement is necessary for schools to be able to reach parents to help improve their and their child's mental health



Step number	Scale and effect (parameter) entered	Number effects in	PRESS	F Value	Pr > F
	SIES-RP				
0	Intercept	1	52.3	0.00	1.0000
1	Psychological distress score (measured as Kessler 10)	2	49.92	12.45	0.0005
2	School C	3	47.78	11.30	0.0009
3	School D	4	46.67	17.05	< 0.0001
4	Current visa category in Australia	5	43.73	6.70	0.0103
5	Duration of living in Australia	6	41.68*	12.59	0.0005
	SCES-RP				
0	Intercept	1	97.11	0.00	1.000
1	Number of PMLDs	2	93.57	10.44	0.0014
2	English proficiency	3	91.06	7.63	0.0062
3	Employment status	4	90.13*	3.99	0.0471
	SBS-RP				
0	Intercept	1	38.37	0.00	1.000
1	Psychological distress score (measured as)	2	35.45	21.59	< 0.0001
2	Number of PMLDs	3	34.80	5.83	0.0166
3	School D	4	34.28	4.59	0.0333
4	Highest level of education	5	33.72*	5.62	0.0186
	CBRS				
0	Intercept	1	49.47	0.00	1.000
1	Duration of living in Australia	2	46.55	14.79	0.0002
2	Highest level of education	3	45.29	7.97	0.0052
3	Employment status	4	44.80*	4.18	0.0423

Table 7. LASSO regression calculation: Forward selection summary for each of the outcome variable.

*PRESS, predicted residual sum of squares statistic (leave-one-out cross-validation).

CBRS, Cultural Broker Relationship Scale; LASSO, least absolute shrinkage and selection operator; PMLD, postmigration living difficulty; SBS-RP, School Belonging Scale-Refugee Parent; SCES-RP, School Community Engagement Scale-Refugee Parent; SIES-RP, School Internal Engagement Scale-Refugee Parent.

https://doi.org/10.1371/journal.pmed.1003512.t007

[60,61]. Lower parental engagement may be one symptom of the more general finding that poor parent mental health, particularly among female caregivers, is a risk factor for negative child outcomes inclusive of refugee and war-affected communities [62–65]. Nearly half of the current parent sample gave indications of a mild to severe mental health disorder. The high rates of mental health issues seen in the refugee population relative to native-born populations may be why psychological distress emerged as a particular risk factor for refugee parents [66].

It is understandable, too, that parents preoccupied with complex life circumstances may have limited capacity to prioritise school engagement. This is consistent with past research documenting a focus on economic survival as a barrier to engagement [19,20,23,24]. It is interesting that PMLD impacted on parents' more affective sense of school belonging. Based on social and cultural capital theories, one might expect stretched resources to have more impact on the more resource-heavy behavioural or cognitive expressions of school engagement. Alternatively, it could be that stressors that challenge one's sense of belonging within the host country more broadly—such as discrimination, communication difficulties, issues with government services, and fears of being sent home—necessarily impinge then on the sense of belonging within the smaller microsystems of that country, including school.

It was noteworthy that English language was not associated with any of the 4 measures of parent engagement. There is extensive literature citing language as a barrier to school engagement in minority populations [40-45]. However, the schools under study were culturally

diverse (over 90% of families did not speak English as a first language), with Arabic being one of the prominent languages. To this end, the schools were familiar with accommodating language needs perhaps more so than typically less culturally diverse schools.

Viewed collectively, the cultural broker scale findings are encouraging evidence that cultural brokers may be assisting target families most in need, i.e., newly arrived families who are not yet employed. This suggests sage use of school capacity. Families who have lived in Australia for longer and are employed and more educated likely have less need to engage with the school cultural broker, given their likely familiarity with the school and capacity to participate in the community.

However, the finding that the longer a parent had lived in Australia was related to lower internal school engagement is curious, especially given the almost opposite relationship in permanent residency or citizenship status (relative to humanitarian visa status) being associated with higher internal school engagement. One might expect a parent who had lived in Australia for longer to be more socially and culturally resourced and thus more able to engage with the school. One potential explanation, similar to the cultural broker reasoning detailed above, is that schools are so geared towards serving newcomers that long-term residents feel less engaged. The relationship between refugee health and time lived in the resettled country does appear to be a nonlinear one. There is some suggestion of a "honeymoon" period whereby refugee health improves significantly upon arrival in the host country, attributed in part to an acute sense of "relief" and an influx of government support, but once this favourable phase passes and government support ceases, refugee health can worsen [67].

The findings carry authenticity when considered that the refugee families were able to be interviewed in their native language. This diverges from many culturally diverse studies that often require English language proficiency, lending itself to a biased, potentially more acculturated, sample. However, the study encompasses many limitations, not least that the reliability of the measures is compromised with the omission of accredited translators and back translation. Some questionnaire items were also double barrelled and could have been interpreted ambiguously.

The most significant limitation of the study is the cross-sectional design. This hinders any casual interpretation and prevents ruling out any other confounding variable underlying the associations found. Moreover, Arabic-speaking families were almost in the majority in the current participating schools, which limits the generalisability of the findings to other schools where refugee families may subsume more of a minority presence or where the host country's language is dominant. Another weakness is that parental engagement was assessed via only one (parent) informant, and there is a noted tendency for parents to inflate self-reports of their school engagement for understandable desirability motives [68]. Furthermore, the authors are careful that this paper does not assume a responsibility on refugee parents to engage, but rather note that parent engagement is a bidirectional relationship between school and the home.

Future research is required to separate out the nuances of each newly developed measure with distinct measures of validity, including predictive and divergent validity. The sociodemographic findings also need to be replicated in larger, longitudinal samples. For example, the number of employed parents in the study was very small, so any interpretation regarding this factor needs to be treated with caution. The small sample sizes across a wide duration of stay from 3 months to 21 years also unfortunately prohibited comparisons between recently arrived and long-term residents in Australia. This would be useful in interpreting the curious finding that the longer a parent had lived in Australia was related to lower internal school engagement.

Limitations of the emerging psychometrics acknowledged, it is hoped that the developed tools will enable a more targeted robust evaluation of the impact of school initiatives on refugee parent engagement, knowing how vital this is in relation to child outcomes. For example,



this paper now provides a measure with which to quantitatively evaluate the quality of the important relationship between the school-based bilingual cultural broker and parent [6,23,26].

The current study forms the baseline data of a larger study that follows parent and child dyads from the last years of primary school across the transition into secondary school. This will enable us to explore the direction and magnitude of change in the different dimensions of parent engagement, and, importantly, meaningful longitudinal associations between different dimensions of parent engagement and child outcomes. Individual item predictor analysis would also be helpful in guiding interventions and pinpointing exactly which tangible dimensions of engagement schools should be focusing on. For instance, during the scale development, it was pertinent that items related to effective communication correlated strongly with parent engagement items, and items related to social capital correlated strongly with acculturation items. Consideration of online parent connectivity in terms of digital literacy, devices, and network may also be a fruitful future endeavour.

Strong family systems are an important protective factor in optimising refugee children's well-being [34]. In terms of enhancing parent school engagement among refugee families, the study findings support the implementation of initiatives that consider the identification and capacity building of parents who are experiencing psychological distress and/or struggling with resettlement stressors. This invites improvements at both government and school levels. Within the school system, a refugee model of engagement that includes the employment of a bilingual cultural broker, in combination with parent outreach programs, is one seemingly positive way through which to provide targeted support to unemployed newly arrived refugee families [6,23,26,31]. At the macro level, the federal system needs to consider policies that minimise the economic and psychosocial challenges that may interfere with refugee parents' capacity to engage with school.

Supporting information

S1 Prospective Protocol. Prospective protocol based on approved ethics submission (HC15833).

(DOCX)

S1 STROBE Checklist. Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) Checklist.

(DOCX)

S1 Data Tables. Pertaining to the development and psychometrics of study scales. (DOCX)

Acknowledgments

The authors would like to thank the supporting schools, all the parents who participated, and research team members, Sajia Faiz and Holya Hassan.

Author Contributions

Conceptualization: Jess R. Baker, Derrick Silove, Susan Rees.

Data curation: Deserae Horswood.

Formal analysis: Mohammed Mohsin.

Funding acquisition: Derrick Silove, Susan Rees.



Investigation: Jess R. Baker, Deserae Horswood, Afaf Al-Shammari.

Methodology: Jess R. Baker, Derrick Silove, Deserae Horswood.

Project administration: Jess R. Baker, Deserae Horswood, Afaf Al-Shammari.

Resources: Jess R. Baker, Deserae Horswood, Afaf Al-Shammari.

Supervision: Derrick Silove, Valsamma Eapen.

- Writing original draft: Jess R. Baker, Mohammed Mohsin.
- Writing review & editing: Jess R. Baker, Derrick Silove, Deserae Horswood, Afaf Al-Shammari, Mohammed Mohsin, Susan Rees, Valsamma Eapen.

References

- 1. UNHCR. Global Trends: Forced Displacement in 2017. Geneva, Swizterland: UNHCR; 2018.
- 2. UNHCR. The UNHCR Global Report 1999. New York; 2000.
- O'Toole Thommessen SA, Todd BK. How do refugee children experience their new situation in England and Denmark? Implications for educational policy and practice. Child Youth Serv Rev. 2018; 85:228– 38. https://doi.org/10.1016/j.childyouth.2017.12.025
- 4. Fazel M. A moment of change: Facilitating refugee children's mental health in UK schools. Int J Educ Dev. 2015; 41:255–61. https://doi.org/10.1016/j.ijedudev.2014.12.006
- Matthews J. Schooling and settlement: Refugee education in Australia. Int Stud Sociol Educ. 2008; 18 (1):31–45. https://doi.org/10.1080/09620210802195947
- 6. McBrien JL. Educational needs and barriers for refugee students in the united states: A review of the literature. Rev Educ Res. 2005; 75(3):329–64.
- 7. Barnard WM. Parent involvement in elementary school and educational attainment. Child Youth Serv Rev. 2004; 26(1):39–62. https://doi.org/10.1016/j.childyouth.2003.11.002
- Miedel WT, Reynolds AJ. Parent Involvement in Early Intervention for Disadvantaged Children: Does It Matter? J Sch Psychol. 1999; 37(4):379–402. https://doi.org/10.1016/S0022-4405(99)00023-0
- 9. Marcon RA. Positive relationships between parent school involvement and public school inner-city preschoolers' development and academic performance. School Psych Rev. 1999; 28(3):395–412.
- Izzo CV, Weissberg RP, Kasprow WJ, Fendrich MA. Longitudinal Assessment of Teacher Perceptions of Parent Involvement in Children's Education and School Performance. Am J Community Psychol. 1999; 27 (6):817–39. https://doi.org/10.1023/a:1022262625984 PMID: 10723536
- Hill NE, Castellino DR, Lansford JE, Nowlin P, Dodge KA, Bates JE, et al. Parent academic involvement as related to school behavior, achievement, and aspirations: demographic variations across adolescence. Child Dev. 2004; 75(5):1491–509. <u>https://doi.org/10.1111/j.1467-8624.2004.00753.x</u> PMID: 15369527.
- 12. Jeynes WH. Parental involvement research: Moving to the next level. The School Community Journal. 2011; 21(1):9–18.
- Jeynes WHA. Meta-Analysis: The Effects of Parental Involvement on Minority Children's Academic Achievement. Educ Urban Soc. 2003; 35(2):202–18. https://doi.org/10.1177/0013124502239392
- 14. Fan X, Chen M. Parental Involvement and Students' Academic Achievement: A Meta-Analysis. Educ Psychol Rev. 2001; 13:1–22. https://doi.org/10.1023/A:1009048817385
- Grolnick WS, Slowiaczek ML. Parents' Involvement in Children's Schooling: A Multidimensional Conceptualization and Motivational Model. Child Dev. 1994; 65(1):237–52. https://doi.org/10.1111/j. 1467-8624.1994.tb00747.x PMID: 8131650
- Epstein JL, Sanders MG, Simon BS, Salinas KC, Jansorn NR, Van Voorhis FL. School, family, and community partnerships: Your handbook for action, 2nd ed. Thousand Oaks, CA, US: Corwin Press; 2002. xi, 379–xi, p.
- Ho E. Effects of Parental Involvement On Eight Grade Achievement. Sociol Educ. 1996;69. <u>https://doi.org/10.2307/2112802</u>
- Bower HA, Griffin D. Can the Epstein Model of Parental Involvement Work in a High-Minority, High-Poverty Elementary School? A Case Study. Prof Sch Couns. 2011; 15(2):2156759X1101500201. https://doi.org/10.1177/2156759x1101500201



- Miller E, Ziaian T, Esterman A. Australian school practices and the education experiences of students with a refugee background: a review of the literature. Int J Incl Educ. 2018; 22(4):339–59. <u>https://doi.org/10.1080/13603116.2017.1365955</u>
- Ibrahim HH, Small D, Grimley M. Parent/school interface: Current communication practices and their implication for Somali parents. N Z J Educ Stud. 2009; 44(2):19–30.
- Hughes N, Beirens H. Enhancing Educational Support: Towards Holistic, Responsive and Strengthbased Services for Young Refugees and Asylum-seekers. Child Soc. 2007; 21:261–72. https://doi.org/ 10.1111/j.1099-0860.2007.00098.x
- Blackmore J, Hutchison K. Ambivalent relations: the 'tricky footwork' of parental involvement in school communities. Int J Incl Educ. 2010; 14(5):499–515. https://doi.org/10.1080/13603110802657685
- Rah Y, Choi S, Nguyễn TST. Building bridges between refugee parents and schools. Int J Leadersh Educ. 2009; 12(4):347–65. https://doi.org/10.1080/13603120802609867
- 24. Collignon FF, Men M, Tan S. Finding Ways In: Community-Based Perspectives on Southeast Asian Family Involvement With Schools in a New England State. Journal of Education for Students Placed at Risk (JESPAR). 2001; 6(1–2):27–44. https://doi.org/10.1207/S15327671ESPR0601-2_3
- 25. UNHCR. Turn the Tide: Refugee Education in Crisis. Geneva, Switzerland: UNHCR, 2018.
- McBrien JL. The importance of context: Vietnamese, Somali, and Iranian refugee mothers discuss their resettled lives and involvement in their children's schools. Compare: A Journal of Comparative and International Education. 2011; 41(1):75–90. https://doi.org/10.1080/03057925.2010.523168
- Mau W-C. Parental influences on the high school students' academic achievement: A comparison of Asian immigrants, Asian Americans, and White Americans. Psychol Sch. 1997; 34(3):267–77. <u>https:// doi.org/10.1002/(sici)1520-6807(199707)34:3<267::aid-pits9>3.0.co;2-l
 </u>
- Desimone L. Linking Parent Involvement with Student Achievement: Do Race and Income Matter? J Educ Res. 1999; 93(1):11–30.
- Fazel M, Reed RV, Panter-Brick C, Stein A. Mental health of displaced and refugee children resettled in high-income countries: risk and protective factors. Lancet. 2012; 379(9812):266–82. https://doi.org/10. 1016/S0140-6736(11)60051-2 PMID: 21835459
- **30.** Hynes T. The issue of 'trust' or 'mistrust' in research with refugees: choices, caveats and considerations for researchers. Geneva: UNHCR, Evaluation and Policy Analysis Unit, 2003.
- Block K, Cross S, Riggs E, Gibbs L. Supporting schools to create an inclusive environment for refugee students. Int J Incl Educ. 2014; 18(12):1337–55. https://doi.org/10.1080/13603116.2014. 899636
- Naidoo L. Developing social inclusion through after-school homework tutoring: a study of African refugee students in Greater Western Sydney. Br J Sociol Educ. 2009; 30(3):261–73. <u>https://doi.org/10.1080/01425690902812547</u>
- Watkins M, Noble G, Wong A. IT'S COMPLEX! Working with Students of Refugee Backgrounds and their Families in New South Wales Public Schools, New South Wales Teachers Federation 2019.
- Baker JR, Raman S, Kohlhoff J, George A, Kaplun C, Dadich A, et al. Optimising refugee children's health/wellbeing in preparation for primary and secondary school: a qualitative inquiry. BMC Public Health. 2019; 19(1):812. https://doi.org/10.1186/s12889-019-7183-5 PMID: 31242897
- Kim Y. Minority parental involvement and school barriers: Moving the focus away from deficiencies of parents. Educ Res Rev. 2009; 4(2):80–102. https://doi.org/10.1016/j.edurev.2009.02.003
- Brar-Josan N, Yohani SC. Cultural brokers' role in facilitating informal and formal mental health supports for refugee youth in school and community context: a Canadian case study. Br J Guid Counc. 2019; 47 (4):512–23. https://doi.org/10.1080/03069885.2017.1403010
- Povey J, Campbell AK, Willis L-D, Haynes M, Western M, Bennett S, et al. Engaging parents in schools and building parent-school partnerships: The role of school and parent organisation leadership. Int J Educ Res. 2016; 79:128–41. https://doi.org/10.1016/i.ijer.2016.07.005
- Murray KW, Finigan-Carr N, Jones V, Copeland-Linder N, Haynie DL, Cheng TL. Barriers and Facilitators to School-Based Parent Involvement for Parents of Urban Public Middle School Students. SAGE Open. 2014; 4(4):10.1177/2158244014558030. Epub 2014/11/18. https://doi.org/10.1177/ 2158244014558030 PMID: 27088049.
- Rattenborg K, MacPhee D, Walker AK, Miller-Heyl J. Pathways to Parental Engagement: Contributions of Parents, Teachers, and Schools in Cultural Context. Early Educ Dev. 2019; 30(3):315–36. <u>https://doi.org/10.1080/10409289.2018.1526577</u>
- Ramirez AYF. Dismay and Disappointment: Parental Involvement of Latino Immigrant Parents. Urban Rev. 2003; 35(2):93–110. https://doi.org/10.1023/a:1023705511946



- García Coll C, Akiba D, Palacios N, Bailey B, Silver R, DiMartino L, et al. Parental Involvement in Children's Education: Lessons from Three Immigrant Groups. Parenting. 2002; 2(3):303–24. https://doi.org/10.1207/S15327922PAR0203_05
- Sohn S, Wang XC. Immigrant parents' involvement in American schools: Perspectives from Korean mothers. Early Childhood Educ J. 2006; 34(2):125–32. https://doi.org/10.1007/s10643-006-0070-6
- **43.** Peña DC. Parent Involvement: Influencing Factors and Implications. J Educ Res. 2000; 94(1):42–54. https://doi.org/10.1080/00220670009598741
- Huss-Keeler RL. Teacher perception of ethnic and linguistic minority parental involvement and its relationships to children's language and literacy learning: A case study. Teach Teach Educ. 1997; 13 (2):171–82. https://doi.org/10.1016/S0742-051X(96)00018-2
- Wong SW, Hughes JN. Ethnicity and Language Contributions to Dimensions of Parent Involvement. School Psych Rev 2006; 35(4):645–62. PMID: 18084632.
- Turney K, Kao G. Barriers to School Involvement: Are Immigrant Parents Disadvantaged? J Educ Res. 2009; 102(4):257–71. https://doi.org/10.3200/JOER.102.4.257–271
- 47. Department of Social Services. Settlement Data Reports January to December 2016 http://data.gov.au/ dataset/8d1b90a9-a4d7-4b10-ad6a-8273722c8628/resource/7fc5da75-0d24-4ec6-ad89-891aed7244ca/download/settlement-data-reports-january-to-december-2016.xlsx
- Jimerson SR, Campos E, Greif JL. Toward an Understanding of Definitions and Measures of School Engagement and Related Terms. Calif Sch Psychol. 2003; 8(1):7–27. https://doi.org/10.1007/ bf03340893
- 49. Horswood D, Baker J, Fazel M, Rees S, Heslop L, Silove D. School factors related to the emotional wellbeing and resettlement outcomes of students from refugee backgrounds: protocol for a systematic review. Syst Rev. 2019; 8(1):107. https://doi.org/10.1186/s13643-019-1016-6 PMID: 31039825
- Goodenow C. The psychological sense of school membership among adolescents: Scale development and educational correlates. Psychol Sch. 1993; 30(1):79–90. https://doi.org/10.1002/1520-6807 (199301)30:1<79::Aid-pits2310300113>3.0.Co;2-x
- 51. Muthén LK, Muthén BO. Mplus User's Guide. 5th2007.
- Bear GG, Yang C, Pasipanodya E. Assessing School Climate:Validation of a Brief Measure of the Perceptions of Parents. J Psychoeduc Assess. 2015; 33(2):115–29. <u>https://doi.org/10.1177/</u> 0734282914545748
- 53. Slade T, Johnston A, Browne MAO, Andrews G, Whiteford H. 2007 National Survey of Mental Health and Wellbeing: Methods and key findings. Aust NZ J Psychiatry. 2009; 43(7):594–605. https://doi.org/ 10.1080/00048670902970882 PubMed PMID: ISI:000267012100003. PMID: 19530016
- Sulaiman-Hill CM, Thompson SC. Selecting instruments for assessing psychological wellbeing in Afghan and Kurdish refugee groups. BMC Res Notes. 2010; 3:237. Epub 2010/09/10. <u>https://doi.org/ 10.1186/1756-0500-3-237 PMID: 20825681; PubMed Central PMCID: PMC2949661.</u>
- Schweitzer R, Melville F, Steel Z, Lacherez P. Trauma, Post-Migration Living Difficulties, and Social Support as Predictors of Psychological Adjustment in Resettled Sudanese Refugees. Aust N Z J Psychiatry. 2006; 40(2):179–87. https://doi.org/10.1080/j.1440-1614.2006.01766.x PMID: 16476137
- Aragona M, Pucci D, Mazzetti M, Maisano B, Geraci S. Traumatic events, post-migration living difficulties and post-traumatic symptoms in first generation immigrants: a primary care study. Ann Ist Super Sanita. 2013; 49(2):169–75. Epub 2013/06/19. https://doi.org/10.4415/ANN_13_02_08 PMID: 23771261.
- Kleijn WC, Hovens JE, Rodenburg JJ. Posttraumatic Stress Symptoms in Refugees: Assessments with the Harvard Trauma Questionnaire and the Hopkins Symptom Checklist–25 in Different Languages. Psychol Rep. 2001; 88(2):527–32. https://doi.org/10.2466/pr0.2001.88.2.527 PMID: 11351903
- Sigvardsdotter E, Malm A, Tinghög P, Vaez M, Saboonchi F. Refugee trauma measurement: a review of existing checklists. Public Health Rev. 2016; 37(1):10. <u>https://doi.org/10.1186/s40985-016-0024-5</u> PMID: 29450052
- 59. IBM Corporation. IBM SPSS Statistics for Windows. 25.0 ed. Armonk, NY: IBM Corp.; 2017.
- Kia-Keating M, Ellis BH. Belonging and connection to school in resettlement: Young refugees, school belonging, and psychosocial adjustment. Clin Child Psychol Psychiatry. 2007; 12(1):29–43. <u>https://doi.org/10.1177/1359104507071052</u> PMID: 17375808
- McNeely CA, Morland L, Doty SB, Meschke LL, Awad S, Husain A, et al. How Schools Can Promote Healthy Development for Newly Arrived Immigrant and Refugee Adolescents: Research Priorities. J Sch Health. 2017; 87(2):121–32. https://doi.org/10.1111/josh.12477 PMID: 28076923
- **62.** Ekblad S. Psychosocial adaptation of children while housed in a Swedish refugee camp: Aftermath of the collapse of Yugoslavia. STRESS MED. 1993; 9(3):159–66.



- 63. Fox PG, Cowell JM, Montgomery AC. Southeast Asian refugee children: violence experience and depression. Int J Psychiatr Nurs Res. 1999; 5(2):589–600. PMID: 10734851
- Thabet AA, Ibraheem AN, Shivram R, Winter EA, Vostanis P. Parenting support and PTSD in children of a war zone. Int J Soc Psychiatry. 2009; 55(3):226–37. <u>https://doi.org/10.1177/0020764008096100</u> PMID: 19383666
- Betancourt TS, Khan KT. The mental health of children affected by armed conflict: Protective processes and pathways to resilience. Int Rev Psychiatry. 2008; 20(3):317–28. <u>https://doi.org/10.1080/</u> 09540260802090363 PMID: 18569183
- McFarlane CA, Kaplan I, Lawrence JA. Psychosocial indicators of wellbeing for resettled refugee children and youth: Conceptual and developmental directions. Child Indic Res. 2011; 4(4):647–77. https://doi.org/10.1007/s12187-010-9100-4
- Uribe Guajardo MG, Slewa-Younan S, Smith M, Eagar S, Stone G. Psychological distress is influenced by length of stay in resettled Iraqi refugees in Australia. Int J Ment Heal Syst. 2016; 10:4. <u>https://doi.org/ 10.1186/s13033-016-0036-z PMID: 26793271</u>.
- **68.** Nord CW, National Center for Education S. Home literacy activities and signs of children's emerging literacy, 1993 and 1999: Washington, DC: National Center for Education Statistics; 2000.



© 2021 Baker et al. This is an open access article distributed under the terms of the Creative Commons Attribution License:

http://creativecommons.org/licenses/by/4.0/(the "License"), which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited. Notwithstanding the ProQuest Terms and Conditions, you may use this content in accordance with the terms of the License.

